Author’s response to reviews

Title: Breast metastasis from a squamous cells carcinoma of the oropharynx: description of a case report

Authors:

Raffaele Longo (raflongo@libero.it)
Emmanuelle Melgar (Emmanuelle-melgar@hotmail.fr)
Marco Campitiello (m.campitiello@chr-metz-thionville.fr)
Francesca Plastino (f.plastino@chr-metz-thionville.fr)
Nada Eid (n.eid@chr-metz-thionville.fr)
Isabelle Quirin (i.quirin@chr-metz-thionville.fr)
Laurent Hennequin (l.hennequin@chr-metz-thionville.fr)
Yves Grignon (y.grignon@chr-metz-thionville.fr)
Michel Gunther (m.gunther@chr-metz-thionville.fr)
Philippe Quetin (p.quetin@chr-metz-thionville.fr)

Version: 2 Date: 24 Oct 2017

Author’s response to reviews:

JMCR-D-17-00400

Breast metastasis from a squamous cells carcinoma of the oropharynx: description of a case report

Raffaele Longo, M.D.; Emmanuelle Melgar; Marco Campitiello; Francesca Plastino; Nada Eid; Isabelle Quirin; Laurent Hennequin; Yves Grignon; Michel Gunther; Philippe Quetin

Journal of Medical Case Reports
Dear Editor,

Please find the final changes of the m/s as suggested.

I would like to thank the Reviewers for their comments.

Reviewer reports:

The manuscript requires an additional revision.

At the end of the Introduction there is a need to add a paragraph that explains why this case report is presented (what is unique and adds to the medical knowledge)

   R: A paragraph was included as suggested

More information is needed:

Give complete past medical, social, family, and environmental history. What medications was the patient on prior to diagnosis?

Give detailed physical and neurological examination on admission. What was the temperature, pulse, blood pressure and temperature, on admission?

   R: A more complete and detailed information on the past patient’s history and physical examination was included in the m/s.

Give all results of laboratory findings (i.e. CBC, liver and renal functions), serology, microbiology etc)

Give information about follow-up for at least 6 months.

   R: Laboratory findings and information on the follow-up were detailed in the m/s.
Was an autopsy performed, and what were the findings?

R: The autopsy was not performed.

In the Discussion – Start with summarizing the case and describe what is unique in this case compared to what is available in the literature.

R: A summary paragraph was included in the Discussion section as suggested.

Yours sincerely,

Dr R. Longo